Author's response to reviews

Title: Need for prolonged immunosuppressive therapy in CLIPPERS - a case report.

Authors:

   Juerd Wijntjes (j.wijntjes@slaz.nl)
   Ernest J. Wouda (e.wouda@slaz.nl)
   Carl E.H. Siegert (c.siegert@slaz.nl)
   Giorgos B. Karas (g.karas@slaz.nl)
   Annemarie M.M. Vlaar (a.vlaar@slaz.nl)

Version: 4 Date: 25 March 2013

Author's response to reviews:

Revisions in response to reviewer Jay-jiguang Zhu

- Besides myself also two of the co authors checked the article for the wrong use terms and spelling mistakes.
- as suggested we changed among others summary into Summary.